Author's response to reviews

Title: Chondroblastoma of the distal femur resected through a small fenestra under CT navigation and endoscope: a case report

Authors:

Tsuyoshi Miyazaki (mtuyo@u-fukui.ac.jp)
Kenzo Uchida (kuchida@u-fukui.ac.jp)
Takafumi Yayama (yayama@u-fukui.ac.jp)
Hideaki Nakajima (nhideaki@u-fukui.ac.jp)
Kazuya Honjoh (kazuya@u-fukui.ac.jp)
Hiroshi Ito (mtuyo2008@yahoo.co.jp)
Yoshinao Oda (mtuyo@u-fukui.ac.jp)
Hisatoshi Baba (mikan@u-fukui.ac.jp)

Version: 2 Date: 27 November 2012

Author's response to reviews: see over
Dear Michael Kidd AM,

Please find our manuscript entitled “Chondroblastoma of the distal femur resected through a small fenestra under CT navigation and endoscope: a case report” by Tsuyoshi Miyazaki et al, for consideration of publication in the Journal of Medical Case reports.

To our best knowledge, this is the first report of the combined use of CT-based navigation system and endoscopy for the treatment of this kind of disease. Since cases of juvenile chondroblastoma is such close relation to the growth plate are difficult to treat due to the disturbance of the bone growth, we believe that this work will significantly advance our understanding for the treatment of this kind of lesions. Due to its specificity, this original case report would be of great interest for a particular clinical specialty of medicine. In accordance to the stated reasons, we believe this manuscript meets the criteria required for “Presentations, management of emerging disease”

Our manuscript has been checked by scientists whose native language is English.

By submitting the manuscript to the Journal of Medical Case reports, the authors understand that the material presented in this paper has not been published before nor has it been submitted for publication to another scientific journal or being considered for publication elsewhere. I attest that this work has been approved by all co-authors. The authors also understand that should the submitted material be accepted for publication in the journal, they will automatically transfer the copyright to the publisher.

I hope that the reviewing process will find the manuscript acceptable for publication in the journal.

Sincerely Yours,

Tsuyoshi Miyazaki, MD, PhD
Department of Orthopaedics and Rehabilitation Medicine
Faculty of Medical Sciences, The University of Fukui
Matsuoka Shimoaizuki 23-3, Eiheiji,
Fukui 910-1193, Japan
email: mtuyo@u-fukui.ac.jp
Phone: +81 (776) 61-8383
Fax: +81 (776) 61-8125